\begin{table}%ta.1
\caption{\label{table:photo-drtau}Photometric flux measurements of DR~Tau.}
%\centerline
 {\small  \begin{tabular}{llr}
\hline\hline
      Wavelength [${\rm \mu m}$] & Flux [Jy] & Ref. \\
\hline
      $0.36	$ & $0.038	\pm 0.021	     $ & { 1} \\
      $0.44	$ & $0.050	\pm 0.049	     $ & { 1} \\
      $0.55	$ & $0.083	\pm 0.060	     $ & { 1} \\
      $0.64	$ & $0.13	\pm 0.06	     $ & { 1} \\
      $0.79	$ & $0.21	\pm 0.01	     $ & { 1} \\
      $1.25	$ & $0.45	\pm 0.01	     $ & { 2} \\
      $1.65	$ & $0.77	\pm 0.04            $ & { 2} \\
      $2.20	$ & $1.13	\pm 0.02            $ & { 2} \\
      $3.50	$ & $2.07	\pm 0.82             $ & { 1} \\
      $3.60 $ & $1.86       \pm 0.20             $ & { 7} \\
      $4.50 $ & $1.89       \pm 0.15             $ & { 7} \\
      $4.80	$ & $2.41	\pm 1.10	     $ & { 1} \\
      $5.80	$ & $1.27	\pm 0.01	     $ & { 3} \\
      $8.00	$ & $1.77	\pm 0.20	     $ & { 7} \\
      $12.0	$ & $3.16	\pm 0.25	     $ & { 4} \\
      $25	$ & $4.30	\pm 0.29	     $ & { 4} \\
      $60	$ & $5.51	\pm 0.54	     $ & { 4} \\
      $100	$ & $6.98	\pm 0.63	     $ & { 4} \\
      $200     $ & $3.98      \pm      0.84	     $ & { 5} \\
      $450	$ & $2.38	\pm 	0.17	     $ & { 6} \\
      $729  $ & $0.40       \pm     0.08        $ & { 8} \\
      $850	$ & $0.53	\pm 	0.01	     $ & { 6} \\
      $1300	$ & $0.16	\pm 	0.01	     $ & { 6} \\
      \hline
    \end{tabular}}
\medskip
References -- 1: Kenyon \& Hartmann~(\cite{kenyon}); 2:~2~MASS catalogue~(Cutri et~al. \cite{cutri}); {3}: Hartmann et~al. (\cite{hartmannIV}); {4}:~Gezari catalogue (\cite{gezari}); {5}:~ISO data archive; {6}:~Andrews \& Williams (\cite{andrewsII}); {7}:~Robitaille et~al. (\cite{robitaille}); {8}:~Beckwith et~al. (\cite{beckwith}).
  \end{table}